Author’s response to reviews

Title: COULD CURRENT FACTORS BE ASSOCIATED WITH RETROSPECTIVE SPORTS INJURIES IN BRAZILIAN JIU-JITSU? A CROSS-SECTIONAL STUDY

Authors:

Dayana das Graças (dayana.dg.fisio@gmail.com)

Letícia Nakamura (leticiahah@hotmail.com)

Fernando Sérgio Silva Barbosa (fernandossb@outlook.com)

Paula Felippe Martinez (paulafmartinez@yahoo.com.br)

Filipe Abdalla Reis (filipe@byofisio.com.br)

Silvio Assis de Oliveira-Junior (oliveirajr.ufms@gmail.com)

Version: 1 Date: 19 Jan 2017

Author’s response to reviews:

January, 19th, 2017To Clare PartridgeBMC Sports Science, Medicine and Rehabilitation http://www.biomedcentral.com/bmcsportsscimedrehabil/ Dear Editor, Thank you very much for the opportunity to resubmit the original paper “Could current factors be associated with retrospective sports injuries in brazilian jiu-jitsu? a cross-sectional purpose” by Dayana das Graças, Letícia Nakamura, Fernando S. S. Barbosa, Paula F. Martinez, Filipe A. dos Reis, and Silvio A. Oliveira-Junior. All comments have been addressed in the reviewed manuscript. We have changed the text as suggested and included some new information as requested. All authors have approved the final submission of the manuscript. No potential conflict of interest relevant to this article was reported. Thank you for your consideration. Sincerely yours, Oliveira-Junior, S. A. Address for Correspondence: School of Physical Therapy, Federal University of Mato Grosso do Sul, s/n, Universitario, 79070-900; Campo Grande, MS, Brazil Phone: 55-67-3345-7837